

Review



Advances in Targeted Therapy for Systemic Lupus Erythematosus: Current Treatments and Novel Approaches

Kazusa Saegusa 🗅, Yumi Tsuchida *🗅, Toshihiko Komai 🔍, Haruka Tsuchiya and Keishi Fujio 🕩

Department of Allergy and Rheumatology, Graduate School of Medicine, The University of Tokyo, 7-3-1 Hongo, Bunkyo-ku, Tokyo 113-8655, Japan; saegusa-kazusa086@g.ecc.u-tokyo.ac.jp (K.S.); komait-int@h.u-tokyo.ac.jp (T.K.); tsuchiyah-int@h.u-tokyo.ac.jp (H.T.); fujiok-int@h.u-tokyo.ac.jp (K.F.) * Correspondence: yumitsuchida@g.ecc.u-tokyo.ac.jp; Tel.: +81-3-3815-5411; Fax: +81-3-3815-5954

Abstract: Systemic lupus erythematosus (SLE) is a chronic autoimmune disease with diverse clinical manifestations that can lead to severe organ damage. The complex pathophysiology of SLE makes treatment selection difficult. This review examines the current evidence for biological therapies in SLE, including the anti-B cell activating factor antibody belimumab; the type I interferon receptor antagonist anifrolumab; the novel calcineurin inhibitor voclosporin; and rituximab, which targets CD20 on B cells. We also describe emerging therapies, including novel agents in development and CD19-directed chimeric antigen receptor (CAR) T cell therapy, which has shown promise in early clinical experience. Recent advances in biomarker research, including interferon signatures and transcriptomic profiles, may facilitate patient stratification and treatment selection. This review offers insights into current and future treatment strategies for patients with SLE by analyzing clinical trial results and recent immunological findings.

Keywords: systemic lupus erythematosus; biologic therapies; targeted therapies; precision medicine

1. Introduction

Systemic lupus erythematosus (SLE) is a chronic autoimmune disease with diverse clinical manifestations and pathophysiological mechanisms. The complex pathophysiology of SLE and heterogeneity among patients make treatment selection difficult.

Immune dysregulation in SLE involves both innate and adaptive immune responses. A loss of immune tolerance leads to the production of autoantibodies against nuclear antigens and other self-antigens. The formation and deposition of immune complexes in various organs trigger inflammation and tissue damage. Key pathogenic features include the excessive production of type I interferon by plasmacytoid dendritic cells, imbalanced T cell responses with impaired regulatory T cell function, B cell hyperactivity driven by elevated levels of B cell activating factor (BAFF), promoting pathogenic autoantibody production, and impaired clearance of apoptotic cells, which provides a source of autoantigens [1].

Even with current standard treatments, including glucocorticoids and immunosuppressants, a recent analysis of the international Definition Of Remission In SLE (DORIS) cohort demonstrated that only 50% of 1652 patients achieved remission or low disease activity [2]. Although glucocorticoids are essential for rapid disease control, their toxicity profile presents a major concern in clinical practice. The short-term adverse effects include an increased risk of serious infections and early mortality [3], while long-term use leads to cumulative organ damage, including osteonecrosis, osteoporosis, and cardiovascular complications [4,5], with damage correlating with glucocorticoid dose and duration [6].



Academic Editor: Joan Torras-Ambròs

Received: 28 December 2024 Revised: 18 January 2025 Accepted: 21 January 2025 Published: 23 January 2025

Citation: Saegusa, K.; Tsuchida, Y.; Komai, T.; Tsuchiya, H.; Fujio, K. Advances in Targeted Therapy for Systemic Lupus Erythematosus: Current Treatments and Novel Approaches. *Int. J. Mol. Sci.* **2025**, *26*, 929. https://doi.org/10.3390/ ijms26030929

Copyright: © 2025 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/ licenses/by/4.0/). The therapeutic approach to SLE has changed significantly over the past decade, shifting from conventional immunosuppression to targeted therapies that aim to reduce both disease activity and treatment-related damage. The currently available approved targeted therapies for SLE include belimumab as the initial biological agent, followed by anifrolumab. For lupus nephritis (LN), the novel calcineurin inhibitor voclosporin has gained approval. Although not approved in many countries, rituximab is widely used in clinical practice, particularly for refractory cases. Recent recommendations from the European League Against Rheumatism (EULAR) [7] and Kidney Disease: Improving Global Outcomes (KDIGO) guidelines on LN [8] have increasingly emphasized the importance of these biological agents and the novel calcineurin inhibitor.

Recent clinical data show responses to CD19-directed chimeric antigen receptor (CAR) T cell therapy and T cell engager (TCE) therapy in patients with refractory SLE. Other treatments under investigation include new biological agents, small molecule inhibitors, and cell-based therapies, with a focus on those currently in or with potential for phase III trials. These treatments target specific pathways in SLE pathogenesis, reflecting our current understanding of disease mechanisms (Figure 1).



Figure 1. Therapeutic targets in systemic lupus erythematosus. The figure shows molecular targets of current and emerging SLE therapies. Approved therapies include belimumab (targeting BAFF) and anifrolumab (targeting IFNAR). Emerging therapies target various pathways: litifilimab (BDCA2 on pDCs), rituximab and obinutuzumab (CD20 on B cells), telitacicept (BAFF/APRIL), and dapirolizumab (CD40L on T cells). Janus kinase (JAK) inhibitors upadacitinib and deucravacitinib target JAK1 and TYK2, respectively, modulating cytokine signaling (shown in inset). APC, antigenpresenting cells; APRIL, A proliferation-inducing ligand; BAFF, B cell activating factor; BAFF-R, B cell activating factor receptor; BCMA, B cell maturation antigen; BDCA2, blood dendritic cell antigen 2; IFN, interferon; IFNAR, interferon receptor; JAK1, Janus kinase 1; pDC, plasmacytoid dendritic cells; TACI, Transmembrane Activator and Calcium Modulator and Cyclophilin Ligand Interactor; TYK2, tyrosine kinase 2. Figure prepared with BioRender.

The availability of multiple therapeutic options with different mechanisms requires careful treatment selection. Patient-specific factors, including disease manifestations,

biomarker profiles, and genetic factors, guide treatment decisions. This review examines the clinical evidence for current targeted therapies in SLE, discusses emerging treatments, and considers approaches to treatment selection.

2. Current Landscape of Evidence-Based and Novel Therapies in Systemic Lupus Erythematosus

2.1. Belimumab

Belimumab is a human monoclonal antibody targeting B cell activating factor/B lymphocyte stimulator (BAFF/BLyS). BAFF is a key regulator in B cell survival, differentiation, and maturation. In SLE, excess BAFF expression contributes to autoreactive B cell survival and autoantibody production. Belimumab binds to soluble BAFF, blocking its interaction with B cell receptors and reducing B cell survival.

Two phase III trials established the clinical efficacy of belimumab. BLISS-52 [9] and BLISS-76 [10] enrolled seropositive patients with active SLE (Safety of Estrogens in Lupus Erythematosus National Assessment–SLE Disease Activity Index (SELENA-SLEDAI) score ≥ 6) but excluded those with severe active LN or central nervous system involvement. Both trials showed higher Systemic Lupus Erythematosus Responder Index (SRI-4 [11]—a composite index requiring a \geq four-point improvement in clinical Systemic Lupus Erythematosus Disease Activity Index 2000 (SLEDAI-2K)—score, no new severe organ involvement evaluated with the British Isles Lupus Assessment Group (BILAG) index, no worsening in the Physician's Global Assessment (PGA) by ≥ 0.3 -point response rates at week 52 with belimumab 10 mg/kg compared to placebo (BLISS-52: 58% vs. 44%; BLISS-76: 43.2% vs. 33.5%), and reduced severe flare rates (BLISS-52: 14% vs. 23%; BLISS-76: 20.5% vs. 26.5%). The safety profiles of belimumab, including the rates of serious infections, malignancies, and death, were similar between the belimumab and placebo groups. Subsequent analyses identified high disease activity, positive anti-dsDNA antibodies, and low complement levels as response predictors [12].

The BLISS-LN phase III trial [13] evaluated belimumab for active LN, a condition that had been excluded from the earlier BLISS-52 and BLISS-76 trials. At week 104, belimumab added to standard therapy improved both primary efficacy renal response (43% vs. 32%) and complete renal response (30% vs. 20%). Rates of adverse events were comparable between the two groups across 104 weeks. A post-hoc analysis demonstrated a 55% reduction in renal flares compared with standard-of-care alone [14].

Clinical experience with belimumab, including the 10-year OBSErve cohort [15], shows sustained improvements in disease activity and reduced organ damage. A Cochrane systematic review [16] of six randomized controlled trials further supported belimumab's efficacy, demonstrating improved disease activity and glucocorticoid-sparing effects compared to placebo, with no significant differences in rates of serious adverse events, serious infections, or mortality between belimumab and placebo.

A multicenter observational study in Italy revealed that the initiation of belimumab in early disease stages was associated with improved clinical outcomes [17]. The key predictors of treatment response included baseline SLEDAI-2K scores \geq 10, disease duration \leq 2 years, and minimal organ damage. The analysis demonstrated consistent SRI-4 response rates ranging from 49.2% to 66.7% throughout 48 months of follow-up, accompanied by a marked decrease in disease flares. These findings suggest that patients with limited organ damage and high disease activity will achieve optimal therapeutic benefits from belimumab.

The EULAR recommendations [7] support adding belimumab to standard-of-care in patients who do not respond to hydroxychloroquine or cannot reduce glucocorticoids to acceptable maintenance doses. For active proliferative LN, belimumab can be combined

with glucocorticoids and either mycophenolate or low-dose intravenous cyclophosphamide. The KDIGO guideline [8] also includes belimumab with mycophenolate or low-dose intravenous cyclophosphamide among four first-line options, particularly for patients with high flare risk or progressive kidney disease.

2.2. Anifrolumab

Anifrolumab is a human monoclonal antibody that targets the type I interferon receptor 1 (IFNAR1). By blocking IFNAR1, anifrolumab inhibits signaling of all type I interferons, including IFN α , IFN β , and IFN ω . Type I interferons play a central role in the pathogenesis of SLE [18]. A significant proportion of patients with SLE show elevated interferon-regulated gene expression, characterized by the interferon gene signature (IGS).

The efficacy of anifrolumab was investigated in two phase III trials [19]. TULIP-1 [20] did not meet its primary endpoint using the SRI-4, possibly due to the limitations of the SRI-4 in capturing subtle improvements in disease activity. In contrast, TULIP-2 [21] demonstrated the superiority of anifrolumab over placebo using the British Isles Lupus Assessment Group-based Composite Lupus Assessment (BICLA [22], a composite index based on the BILAG-2004 (an updated version of BILAG, incorporating nine organ systems and refined scoring to improve sensitivity), requiring improvement in all BILAG A or B scores at baseline, no new BILAG A or B scores, stable SLEDAI-2K, and no worsening in the PGA (defined as an increase of <0.3 points on a three-point scale) at week 52 (47.8% vs. 31.5%). Both trials showed significant improvements in skin disease and reduced oral glucocorticoid use. In TULIP-2, patients with high IGS showed a higher treatment response rate (48.0% vs. 30.7%) compared with those with low signatures (46.7% vs. 35.5%). Herpes zoster (HZ) occurred more frequently with anifrolumab than with placebo (7.2% vs. 1.1%). A post-hoc analysis of the pooled TULIP trials demonstrated that anifrolumab-treated patients achieved the Lupus Low Disease Activity State (LLDAS) more frequently than placebo (30.0% vs. 19.6%) at week 52 [23].

The MUSE phase IIb study showed significant improvements in rash (SLEDAI-2K resolution: 44.3% vs. 14.8%) and arthritis (SLEDAI-2K improvement: 56.7% vs. 42.4%) compared to placebo [24], with these benefits maintained in its 3-year open-label extension study [25].

The TULIP 3-year extension study also showed sustained improvement in disease activity scores and lower cumulative glucocorticoid use [26]. Anifrolumab demonstrated an overall acceptable safety profile, with an increased risk of HZ during the initial treatment phase that decreases over time. Most HZ cases were manageable with antiviral therapy, and the rates of serious infections (3.7 vs. 3.6 per 100 patient-years) and serious adverse events (8.5 vs. 11.2 per 100 patient-years) were comparable to those in the placebo group. Due to the timing of the recombinant HZ vaccine approval during the study period, few patients received vaccination. The TULIP-LN phase II trial [27] provided additional evidence for anifrolumab's potential efficacy in LN, with the intensified regimen group showing higher complete renal response rates compared to the placebo group (45.5% vs. 31.1%), with response criteria including the normalization of proteinuria and the stabilization of renal function. The safety profiles were almost similar in both groups, though herpes zoster occurred more frequently in the anifrolumab group (16.7% vs. 8.2%).

The recent EULAR recommendations [7] position anifrolumab as an option for firstline therapy in non-renal SLE, alongside belimumab, without requiring prior failure of conventional immunosuppressants, especially for active skin disease. While the safety profile of anifrolumab is acceptable, careful monitoring of HZ is needed. There is limited evidence for active neuropsychiatric SLE, as such cases were excluded from trials.

2.3. Rituximab

Rituximab is a chimeric monoclonal antibody targeting CD20 on B cells, which spares plasma cells and early B cell precursors. While not approved by the US Food and Drug Administration or the European Medicines Agency, clinical experience supports its use in specific clinical settings, particularly in patients with severe or refractory cases.

Two randomized controlled trials, the EXPLORER phase II/III trial [28] for non-renal SLE and the LUNAR phase III trial [29] for LN, did not meet their primary endpoints. The EXPLORER trial assessed rituximab in moderate-to-severe non-renal SLE but found no difference in clinical response between rituximab and placebo at 52 weeks. In the LUNAR trial, which evaluated rituximab with mycophenolate and corticosteroids in class III/IV LN, overall renal response rates were not significantly different (56.9% vs. 45.8%). However, rituximab-treated patients showed greater reductions in anti-dsDNA antibodies, and none required cyclophosphamide rescue therapy, compared to eight patients in the placebo group.

Registry data support the clinical value of rituximab in specific situations. The British Isles Lupus Assessment Group Biologics Register reported reduced disease activity scores and lower glucocorticoid doses in rituximab-treated patients. Serious infections were reported in 10% of patients, with higher frequency in the first 3 months post-rituximab [30]. The French Autoimmunity and Rituximab registry, with 136 patients with SLE, documented clinical responses in severe manifestations, including LN, neuropsychiatric involvement, and hematologic abnormalities. Severe infections occurred in 9% of patients (6.6/100 patient-years), mainly within 3 months after the last infusion [31]. Although the safety data from registry studies was acceptable, infection risk monitoring is important in the first 3 months after rituximab administration.

For severe autoimmune thrombocytopenia in SLE, rituximab is effective as part of acute treatment along with high-dose glucocorticoids (including intravenous methylprednisolone pulses) and/or intravenous immunoglobulin, and/or high-dose intravenous cyclophosphamide [7]. A multicenter study of 71 adults with SLE-associated immune cytopenia showed its efficacy, with a 91% response rate in patients with immune thrombocytopenia. Severe infections were reported in 4.2% of patients treated with rituximab, with no fatal outcomes [32]. Rituximab can also be considered for maintenance therapy, with alternatives such as azathioprine, mycophenolate, or cyclosporine.

Clinical trials have examined rituximab with belimumab. The CALIBRATE phase II trial [33] tested rituximab induction followed by belimumab in patients with refractory LN, finding similar safety profiles between groups. The BEAT-LUPUS phase II trial [34] compared rituximab followed by belimumab versus rituximab followed by placebo, with higher major clinical response rates at week 52 in the belimumab group (48% vs. 35%). The trial identified baseline serum IgA2 anti-dsDNA antibodies as a potential response biomarker, showing a 48% higher response rate in patients with elevated baseline levels. The BLISS-BELIEVE phase III trial [35] compared subcutaneous belimumab with rituximab (BEL/RTX) versus belimumab with placebo (BEL/PBO). While BEL/RTX did not increase disease control rates (SLEDAI-2K \leq 2, without immunosuppressants, prednisone \leq 5 mg/day) at week 52 (19.4% vs. 16.7%), these patients maintained disease control longer and showed greater reductions in anti-dsDNA antibodies and B cell numbers. The combination of rituximab with belimumab did not increase adverse events compared to rituximab alone, as demonstrated in both the CALIBRATE and BLISS-BELIEVE trials [33,35]. These results point to the potential of B cell-targeted combination therapy with biomarker guidance for patient selection.

Voclosporin is a novel calcineurin inhibitor structurally modified from cyclosporine. Compared to conventional calcineurin inhibitors, it demonstrates higher potency in inhibiting T cell activation and IL-2 transcription, with more predictability, eliminating the need for therapeutic drug monitoring.

The AURORA-1 phase III trial [36] demonstrated its efficacy in LN as an add-on therapy to mycophenolate and low-dose steroids. At week 52, complete renal response rates were significantly higher with voclosporin versus placebo (41% vs. 23%). Serious adverse events occurred in 21% of patients in both groups, with pneumonia being the most frequent serious infection (4% in both groups). However, the trial did not assess renal biopsy activity and chronicity indices or evaluate responses between new-onset and relapsed LN.

The AURORA-2 extension study [37] provided evidence for long-term safety and efficacy over 3 years. Among 216 patients, 86.1% completed the study with no unexpected safety concerns reported. While glomerular filtration rate (GFR) decreases and hypertension were more frequent with voclosporin (10.3% vs. 5.0% and 8.6% vs. 7.0%, respectively), the mean estimated GFR (eGFR) remained stable and within the normal range. The eGFR slope over 2 years was $-0.2 \text{ mL/min}/1.73 \text{ m}^2$ with voclosporin versus $-5.4 \text{ mL/min}/1.73 \text{ m}^2$ with placebo. Complete renal response rates increased to 50.9% with voclosporin versus 39.0% with control at 3 years. Notably, the AURORA trials demonstrated efficacy with lower glucocorticoid doses (20–25 mg/day prednisone, rapidly tapered to 5 mg by 12 weeks) than traditional regimens. While voclosporin was studied across LN classes, its efficacy in pure class V LN requires further investigation due to limited patient numbers in clinical trials. A limitation of AURORA-2 was that the voclosporin group had higher renal response rates at study entry, potentially affecting the interpretation of long-term outcomes.

The EULAR recommendations [7] note that voclosporin provides a rapid reduction in proteinuria, which may be beneficial in patients with high baseline proteinuria in the nephrotic range. The KDIGO guideline [8] states that voclosporin should be used cautiously in patients with impaired kidney function (eGFR < 45 mL/min/1.73 m²) or with widespread sclerotic changes on kidney biopsy. It also states that treatment continuation for up to 3 years appears safe, as supported by stable GFR levels in long-term studies.

3. Emerging Therapies

3.1. Novel Biological Agents in Development

Ongoing research has yielded new therapeutic approaches for SLE beyond current standard treatments. This section examines selected agents showing particular promise in phase II and III studies (Table 1).

Obinutuzumab

Obinutuzumab, a type II anti-CD20 monoclonal antibody, differs from rituximab in its CD20 binding properties and mechanisms of B cell depletion. In vitro studies demonstrate superior B cell cytotoxicity compared to rituximab through enhanced Fc gamma receptor-mediated effects and direct cell death, maintaining surface localization after CD20 binding [38].

The NOBILITY phase II trial in LN showed that the addition of obinutuzumab to standard therapy achieved higher complete renal response rates compared to placebo at week 52 (35% vs. 23%) and week 104 (41% vs. 23%), with good tolerability except for increased non-serious infusion reactions [39]. Furthermore, in patients with renal or non-renal SLE with secondary non-response to rituximab, obinutuzumab improved disease activity scores and achieved B cell depletion, suggesting its potential role as an alternative

therapy in refractory cases [40]. These results supported the initiation of an ongoing phase III trial.

Dapirolizumab Pegol

Dapirolizumab pegol is a PEGylated anti-CD40L antibody fragment that blocks CD40-CD40L interactions, which are key pathways in T cell-dependent B cell responses and in autoimmune inflammation in SLE. Early attempts to target this pathway with conventional anti-CD40L antibodies showed promise but were halted due to thromboembolic events [41]. Dapirolizumab pegol was designed without an Fc domain to avoid platelet activation while maintaining therapeutic efficacy.

In a phase II trial of patients with moderately to severely active SLE [42], intravenous dapirolizumab pegol (6/24/45 mg/kg) showed clinical and immunological improvements across all doses compared to placebo at 24 weeks, though a clear dose–response relationship was not established. In addition, dapirolizumab pegol demonstrated an acceptable safety profile with no increased thrombotic risk.

The phase III PHOENYCS GO trial [43] demonstrated superior efficacy of dapirolizumab pegol (24 mg/kg) compared to placebo in moderate-to-severe SLE patients. The primary endpoint showed significantly higher BICLA response rates in the treatment group (49.5% vs. 34.6%), along with improved SRI-4 responses (60.1% vs. 41.1%) and reduced severe flares (11.6% vs. 23.4%). While treatment-related adverse events were more frequent with dapirolizumab pegol, serious adverse events were less common, suggesting an acceptable safety profile.

The PEGylated structure of dapirolizumab pegol may limit placental transfer, suggesting potential use during pregnancy pending further safety data.

Telitacicept

As discussed in the belimumab section, B cells play a central role in SLE pathogenesis through autoantibody production and immune dysregulation. Telitacicept is a fusion protein that combines the extracellular domain of the transmembrane activator and calcium modulator and cyclophilin ligand interactor (TACI) receptor with the Fc portion of human IgG1, targeting two key B cell survival factors: BAFF/BlyS and A proliferation-inducing ligand (APRIL). They promote B cell differentiation, maturation, and survival through binding to cell surface receptors, including TACI. Unlike belimumab, which selectively inhibits BlyS, telitacicept blocks both the BlyS and APRIL signaling pathways, potentially offering the complete suppression of B cell-mediated autoimmunity. This dual inhibition may provide more comprehensive control of aberrant B cell responses in SLE.

In a phase IIb randomized controlled trial of 249 patients with active SLE in China [44], subcutaneous telitacicept (80 mg, 160 mg, or 240 mg weekly) demonstrated significantly higher SRI-4 response rates at week 48 across all treatment groups compared to placebo (71.0–75.8% vs. 33.9%). The 240 mg dose showed additional benefits, including gluco-corticoid dose reduction. The safety profile was comparable to placebo, suggesting good tolerability. Further investigations are needed to confirm its efficacy and safety, including larger trials in more diverse populations. Phase III clinical trials are set to be conducted at multiple sites across 15 countries [45].

• Litifilimab

Litifilimab is a humanized IgG1 monoclonal antibody designed to target blood dendritic cell antigen 2 (BDCA2), which is uniquely expressed on plasmacytoid dendritic cells. By binding to BDCA2, litifilimab suppresses the production of type I interferons, cytokines, and chemokines—key mediators in SLE pathogenesis. This mechanism provides a novel approach to modulating the interferon pathway in SLE, distinct from direct cytokine or receptor blockade. In a phase II trial (LILAC) [46], subcutaneous litifilimab (450 mg) demonstrated superiority over placebo in reducing active joint count at 24 weeks in patients with SLE with arthritis and active skin disease; however, the results of most secondary endpoints were not consistent with that of the primary endpoint. The safety profile included cases of herpes zoster and herpes keratitis, warranting further evaluation. Phase III trials are currently in progress to confirm the efficacy and safety of litifilimab.

Upadacitinib

Upadacitinib is a selective Janus kinase 1 (JAK1) inhibitor that blocks signaling through multiple cytokine pathways involved in SLE pathogenesis, including type I and II interferons and the interleukins IL-2, IL-4, IL-6, IL-10, and IL-15.

In the phase II SLEek trial [47], more patients who received upadacitinib 30 mg once daily achieved the SRI-4 response compared with placebo at week 24 (54.8% vs. 37.3%). It also demonstrated improvements in BICLA response and LLDAS and fewer disease flares. The safety profile was consistent with that observed in other approved indications, such as rheumatoid arthritis and psoriatic arthritis. Further investigation of upadacitinib for moderate to severe SLE in phase III trials is under way. It was also evaluated in combination with the BTK inhibitor elsubrutinib (ABBV-599), showing comparable efficacy to upadacitinib monotherapy but requiring additional studies to determine the optimal therapeutic approach.

Deucravacitinib

Deucravacitinib is an orally administered, selective, allosteric inhibitor of tyrosine kinase 2 (TYK2) that targets its regulatory domain and inhibits signaling pathways mediated by type I interferons as well as the interleukins IL-10, IL-12, and IL-23, which are key cytokines in SLE pathogenesis.

In a phase II trial involving 363 patients with active SLE [48], the SRI-4 response rate at week 32 was significantly higher in patients receiving deucravacitinib 3 mg twice daily compared to placebo (58% vs. 34%). The safety profile was generally acceptable, though increased rates of infections and cutaneous events were noted. Further study is needed, including the ongoing Phase III trial [49], to confirm its efficacy and safety.

• Others

Low-dose IL-2 therapy has gained attention as a potential treatment for SLE based on its ability to expand regulatory T cells (Tregs) without activating effector T cells. While the initial results at 12 weeks were not statistically significant in a controlled trial, the follow-up at 24 weeks revealed superior outcomes compared to placebo in both overall response (66% vs. 37%) and nephritis remission (53.85% vs. 16.67%) [50]. Further studies are underway in China [51,52].

Ustekinumab, targeting IL-12/23 p40, demonstrated efficacy in SLE in a phase II trial with higher SRI-4 response rates compared to placebo (62% vs. 33%) [53] but failed to meet its primary endpoint in phase III testing (44% vs. 56% for placebo), resulting in trial discontinuation [54].

Despite encouraging case reports of secukinumab in the treatment of SLE [55,56], clinical trials of this IL-17A inhibitor were unsuccessful [57–59]. Cases of drug-induced lupus after treatment for ankylosing spondylitis and psoriasis with IL-17A inhibitors have also been reported [60,61], highlighting the complex role of IL-17 in lupus pathogenesis.

Agent	Mechanism of Action	Trial Phase	Patient Population	Primary Endpoint (Treatment vs. Placebo)	Key Secondary Outcome
Obinutuzumab [39]	Type II anti-CD20 mAb	II	Active/chronic LN	CRR at week 52: 35% vs. 23%	CRR at week 104: 41% vs. 23%
Dapirolizumab pegol [42]	PEGylated anti-CD40L	Ш	Moderate to severely active SLE, stable LN	Dose-response relationship of BICLA response rates at week 24: none	BICLA response rate: 48.8–54.5% vs. 37.2%
Telitacicept [44]	TACI-Fc fusion protein (BlyS/APRIL inhibitor)	IIb	Active SLE	SRI-4 response rate at week 48: 71.0–75.8% vs. 33.9%	GC dose reduction with 240 mg dose
Litifilimab [46]	Anti-BDCA2	Ш	SLE (SLEDAI-2K \geq 4)	Total number of active joints at week 24: 19.0 ± 8.4 vs. 21.6 ± 8.5	Most secondary endpoints not met
Upadacitinib [47]	JAK1 inhibitor	Ш	Moderate to severely active SLE	SRI-4 response rate and GC dose ≤ 10 mg QD at week 24: 54.8% vs. 37.3%	SRI-4, BICLA, LLDAS response rate at week 48: 45.2% vs. 32.0%, 53.2% vs. 25.3%, 50.0% vs. 24.0%. Overall flares at week 24: 1.9 vs. 2.8.
Deucravatinib [48]	TYK2 inhibitor	Ш	Active SLE	SRI-4 response rate at week 32: 58% vs. 34%	BICLA, CLASI-50, LLDAS response rates, active joint count at week 48: 57.1% vs. 34.4%, 47.3% vs. 25.6%, 36.6% vs. 13.3%, -8.9 vs7.6.

Table 1. Clinical trials of novel biological agents in systemic lupus erythematosus.

Abbreviations: APRIL, A proliferation-inducing ligand; BDCA2, blood dendritic cell antigen 2; BICLA, British Isles Lupus Assessment Group-based Composite Lupus Assessment; BlyS, B cell activating factor; CLASI, Cutaneous Lupus Erythematosus Disease Area and Severity Index; CRR, complete renal response; GC, glucocorticoids; JAK1, Janus kinase 1; LLDAS, Lupus Low Disease Activity State; LN, lupus nephritis; SLE, systemic lupus erythematosus; SLEDAI-2K, Systemic Lupus Erythematosus Disease Activity Index 2000; SRI-4, SLE Responder Index 4; TYK2, tyrosine kinase 2.

3.2. CAR T Cell Therapy and T Cell Engager Therapy

• CAR T Cell Therapy

CD19-directed chimeric antigen receptor (CAR) T cell therapy, originally developed for B cell malignancies, provides a novel approach for severe SLE. B cells are central to SLE pathogenesis; however, conventional B cell-depleting therapies such as rituximab have shown limited efficacy in clinical trials. This is possibly due to the autoreactive B cells in lymphatic organs and inflammatory tissues and the CD20-negative plasma affecting autoantibody production [62]. CAR T cells are engineered to express receptors targeting CD19 on B cells, leading to profound B cell depletion. Although life-threatening adverse effects, such as high-grade cytokine release syndrome (CRS), immune effector cell-associated neurotoxicity syndrome (ICANS), and immune effector cell-associated hemophagocytic lymphohistiocytosis-like syndrome, have been reported in experiences with malignancies, this approach remains promising. It offers the potential to eliminate

10 of 20

disease-driving autoantibodies and promote immune system restoration, particularly in refractory patients [63].

Initial clinical experience with CAR T cell therapy in SLE has yielded promising results. The first reported case was a 20-year-old woman with severe, treatment-refractory SLE who achieved complete clinical and serological remission following CAR T cell infusion [64]. This report was followed by a case series of five patients with refractory SLE, all of whom achieved DORIS remission within 3 months of treatment [65]. A subsequent cohort study of 15 patients with autoimmune diseases, including 8 with SLE, has expanded these observations [66]. Clinical remission continued after B cell reconstitution at approximately 110 days. These reconstituted B cells show a naïve phenotype with non–class-switched B cell receptors, suggesting restoration of B cell tolerance.

CAR T cell therapy safety profiles appear more favorable in SLE than in B cell malignancies. In patients with cancer, life-threatening adverse effects, such as high-grade CRS and immune effector cell-associated hematotoxicity, rarely occur in patients with SLE. This possibly reflects the lower burden of target cells in SLE than in B cell malignancies [63].

Dual-targeting strategies are also being explored. The B cell maturation antigen (BCMA)-CD19 compound CAR approach targets both B cells and plasma cells. Phase I trial data showed patients testing negative for all disease-associated autoantibodies and sustaining medication-free remission for up to 46 months [67].

Several questions remain concerning appropriate patient selection, long-term safety, and durability of response. The treatment requires specialized centers capable of T cell processing and managing potential complications. Further studies, including controlled clinical trials, are needed to establish the role of CAR T cell therapy in severe SLE. Further understanding of post-reconstitution B cell tolerance may guide future therapeutic strategies.

T Cell Engager Therapy

Another emerging approach involves T cell engagers (TCEs). This approach uses bispecific antibodies to simultaneously bind CD3 on T cells and specific antigens on target cells, such as B cell maturation antigen (BCMA) or CD19 on B cells and plasma cells, thereby activating T cells to mediate the targeted destruction of pathogenic cells in autoimmune diseases. Initially developed for the treatment of hematological malignancies, TCEs offer several practical advantages over CAR T therapy: no requirement for complex manufacturing processes or lymphodepleting chemotherapy, lower costs, and easier dose modification based on clinical response. While both TCEs and CAR T therapy share common adverse events including CRS, ICANS, infections, and hematological toxicities, preliminary data suggest better tolerability of TCEs in autoimmune diseases than in cancer treatment [68,69].

Teclistamab, which targets CD3 and BCMA and is used in relapsed or refractory multiple myeloma [70], has demonstrated promise in treating refractory autoimmune diseases, including systemic sclerosis, Sjögren's syndrome, and rheumatoid arthritis (RA) [68]. In a case report of a young woman with refractory SLE treated with teclistamab [69], the patient experienced complete remission through effective B cell and plasma cell depletion. The drug showed a favorable safety and tolerability profile, with manageable adverse events, including low-grade cytokine release syndrome. Based on these encouraging results with BCMA-targeting therapy, other agents in this class warrant investigation. Among them, elranatamab, another CD3-BCMA TCE approved for multiple myeloma, offers potential advantages, including a longer half-life (22 days vs. 15 days) and fixed dosing rather than weight-based administration [71].

Blinatumomab, targeting CD3 and CD19, achieved synovial B cell depletion in twothirds of RA patients resistant to conventional therapy, demonstrating superior tissue penetration compared to rituximab [72]. In this study, treatment led to significant clinical improvement and reduction in autoantibody levels. Both rituximab and CD19-targeted therapies share a common limitation: CD19-negative long-lived plasma cells remain unaffected, leading to persistent autoantibody production in some patients. This limitation highlights the potential therapeutic advantage of CD3-BCMA TCEs in certain cases, as they can target both B cells and plasma cells expressing BCMA. A trial is currently underway for a new CD19/CD3/human serum albumin-targeting TCE in SLE, designed to direct T cells to eliminate B cells, including those with diminished or undetectable CD19 expression, as determined by immunohistochemistry [73,74].

Further studies are needed to establish the efficacy and safety profile of this approach. Key future challenges include determining suitable patients for TCE therapy and establishing appropriate dosing protocols in autoimmune conditions.

4. Future Perspectives: Precision Medicine in Systemic Lupus Erythematosus

Despite the recent advances in treatment discussed above, SLE remains challenging to manage due to its heterogeneous clinical presentation and unpredictable disease course. Current SLE classification relies primarily on clinical manifestations and serological findings, often failing to capture the underlying molecular heterogeneity driving disease activity and organ damage. The concept of precision medicine offers the potential to overcome this limitation in SLE treatment by tailoring therapies to individual patient characteristics and disease mechanisms.

Genetic Factors in SLE Pathogenesis and Patient Stratification

Genetic factors, as well as epigenetic factors, including DNA methylation, histone modifications, and microRNAs, also contribute to this diversity [75]. Genome-wide association studies (GWAS) have identified over 100 SLE risk loci, many of which are involved in immune regulation, interferon signaling, and B cell function [1,75]. Variants in interferon pathway genes, including *IRF5* and *STAT4*, are associated with altered responses to type I interferons. Polymorphisms in genes encoding B cell signaling molecules, such as *BLK* and *BANK1*, have been associated with altered B cell activation and autoantibody production [76,77]. Additionally, defects in genes involved in nucleic acid degradation and sensing, including *DNASE1* and *DNASE1L3*, contribute to disease development in some patients [1,78,79]. Understanding these genetic variations could help identify patients at risk for disease development and guide therapeutic monitoring.

The identification of organ-specific genetic associations has particular relevance for treatment selection. Variants in platelet-derived growth factor receptor alpha (*PDGFRA*) and *HAS2* have been linked to lupus nephritis development [75,80,81]. *ITGAM* variants show an association with cutaneous, joint, and neuropsychiatric manifestations in addition to renal impairment [75,82–84]. These genetic markers may help identify patients most likely to benefit from specific targeted therapies.

Advancements in transcriptomics provide further insights into genetic factors in SLE. In a study of 162 SLE patients and 99 controls, single-cell RNA sequencing of peripheral blood mononuclear cells showed that monocytes had the most prominent type I IGS among eight cell types. Monocyte IGS inversely correlated with naive CD4⁺ T cell counts, and cell-type-specific expression patterns could stratify patients into two molecular subtypes [85]. These genetic associations uncover important aspects of the molecular pathways driving SLE pathogenesis.

Epigenetic modifications also affect gene expression and immune cell function in SLE, with DNA hypomethylation and histone changes modulating CD4⁺ T cell activity [75,86]. MicroRNAs regulate inflammatory pathways and cytokine production, suggesting their

potential as disease biomarkers [75,87]. These findings could guide personalized treatment approaches.

Biomarker-Based Patient Stratification

The heterogeneity of SLE reflects diverse underlying immunological conditions involving various cell types, including T cells, B cells, neutrophils, dendritic cells, and platelets, as well as the dysregulation of cytokines, complement pathways, and other immune mediators [68]. Identifying and validating biomarkers that reflect these diverse pathophysiological processes is needed to stratify patients accurately.

Several potential biomarkers are currently under investigation. The IGS reflects the increased expression of interferon-stimulated genes, a common feature in SLE that is often associated with disease activity and specific organ manifestations [69]. Circulating proteins, such as CXCL10, galectin-9, and sialic acid-binding Ig-like lectin 1 (Siglec-1), are promising surrogate biomarkers of type I interferon for identifying patients at risk of flares or those likely to benefit from interferon-targeted therapies. CXCL10 is a chemokine involved in inflammation and autoimmune diseases [75], with elevated levels in patients with SLE [88], making it a potential marker for disease activity and flares [89], especially in LN and neuropsychiatric symptoms [88,90,91]. Galectin-9 correlates with disease activity and damage and is a potential marker for neuropsychiatric SLE [92]. Siglec-1, a myeloid cell-surface protein, correlates with IGS and renal complications, though not with disease activity [93].

Serum levels of BAFF and APRIL, cytokines that promote B cell survival and differentiation, are also being investigated. As discussed in the belimumab and the telitacicept sections, elevated levels are associated with SLE disease activity and may predict response to therapies targeting these pathways [75,94].

Urinary biomarkers can reflect kidney involvement in SLE. Monocyte chemoattractant protein-1 is elevated in active LN [95]. TNF-like weak inducer of apoptosis is associated with renal inflammation and may predict treatment response [96]. Neutrophil gelatinase-associated lipocalin is predictive of flares in LN [97].

Enzymes associated with lysophospholipid production are also under investigation [98].

Advanced technologies, such as transcriptomic analyses and mass cytometry, enable more detailed immune profiling, potentially revealing novel biomarkers that reflect specific disease endotypes [75]. A recent study from Japan identified distinct transcriptomic signatures associated with disease establishment and exacerbation, offering potential biomarkers for predicting disease course and treatment response [99]. For example, neutrophil-related signatures have been associated with renal involvement [99]. Transcriptomic analyses have provided a comprehensive view of immune-cell–specific gene regulation in SLE, revealing distinct pathways linked to disease phases. Single-cell RNA sequencing has revealed disease activity-dependent transcriptomic changes [85]. Machine learning may help transcriptomic analyses and contribute to patient stratification [100].

Predicting Treatment Response

Predicting which patients are most likely to benefit from a given therapy is a key goal of precision medicine, which is particularly crucial in SLE, where treatment responses can vary widely. For example, patients with high IGS may be more likely to respond to anifrolumab [95].

Further refinement of treatment response prediction may come from understanding the dynamic interplay between genetic variants, gene expression, and environmental factors (context-dependent expression quantitative trait loci or eQTLs) [101]. For example, context-dependent eQTLs affecting genes such as NXF1 in monocytes could influence responses to therapies targeting Toll-like receptor 7 pathways. Another study demonstrated the

13 of 20

potential of disease-activity signatures for predicting treatment response, showing that good responders to belimumab exhibited transcriptomic changes that counteracted these signatures [99]. This suggests that monitoring such signatures could help assess treatment efficacy and guide therapeutic decisions.

Realizing precision medicine in SLE faces several challenges. Further research is needed to standardize these biomarker assays and validate them across diverse populations before widespread adoption.

5. Conclusions

Recent advances in targeted therapies have expanded treatment options for patients with SLE. Clinical evidence supports the efficacy of B cell targeted therapy with belimumab, particularly in patients with high disease activity and positive serological markers. The type I interferon receptor antagonist anifrolumab has shown benefits, especially in patients with active skin disease and high IGS. For LN, voclosporin provides rapid proteinuria reduction with predictable pharmacokinetics. While not approved in many countries, rituximab remains valuable for specific manifestations, including severe cytopenia and refractory disease.

Several promising agents are under investigation, including obinutuzumab, dapirolizumab pegol, and telitacicept, each targeting distinct pathways in SLE pathogenesis. Novel approaches such as dual-targeting strategies and small molecule inhibitors may further expand therapeutic options. Early clinical experiences with CAR T cell therapy and TCE therapy suggest their potential in severe, treatment-resistant cases.

The shift toward precision medicine in SLE treatment requires improved patient stratification based on molecular and clinical characteristics. Research into biomarkers, including interferon signatures and transcriptomic profiles, may help identify patients most likely to benefit from specific therapies. Challenges remain in standardizing these biomarker assays and validating them across diverse populations.

Additional research is required to identify reliable predictors of treatment response, understand treatment resistance mechanisms, and develop strategies for treatment sequencing or combination. A growing understanding of SLE pathogenesis may contribute to realizing tailored therapies to achieve sustained remission while minimizing organ damage.

Author Contributions: K.S. conceived the presented idea and drafted the manuscript. Y.T., T.K., H.T., and K.F. professionally edited the manuscript. All authors have read and agreed to the published version of the manuscript.

Funding: This research was supported by funding from the Center of Innovation Program from Japan Science and Technology Agency (JST) (JPMJCE1304 to K.F.), the Ministry of Education, Culture, Sports, and the Japan Agency for Medical Research and Development (AMED) (JP21tm0424221 to K.F.), the Ministry of Education, Culture, Sports, and the AMED (JP21zf0127004 to K.F.), the Ministry of Education, Culture, Sports, and the AMED (JP22ek0410074 to K.F.), the Ministry of Education, Culture, Sports, and the AMED (JP223fa627001 to K.F.), the Ministry of Education, Culture, Sports, and the AMED (JP233fa627001 to K.F.), the Ministry of Education, Culture, Sports, and the AMED (JP233fa627001 to K.F.), the Ministry of Education, Culture, Sports, and the AMED (JP23gm1810005 to K.F.), Japan Society for the Promotion of Science Grant-in-Aid for Scientific Research (C) (JP22K08559 to Y.T.).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Conflicts of Interest: K.S. declares no conflicts of interest. Y.T. has received speaking fees and/or honoraria from AbbVie, Janssen, GlaxoSmithKline, Astellas Pharma Inc., Daiichi Sankyo, Eli Lilly, Novartis, Eisai, Asahi Kasei Corporation, and AstraZeneca. T.K. has received speaking fees and/or honoraria from Tanabe Mitsubishi, Kissei, Pfizer, Chugai, Daiichi-Sankyo, Eli Lilly, Novartis, Eisai,

Asahi Kasei, AbbVie, Janssen, and received grants and speaking fees from GlaxoSmithKline. H.T. has received speaking fees and/or honoraria from AbbVie, Amgen, Asahi Kasei, Astellas, Bristol-Myers Squibb, Chugai, Daiichi-Sankyo, Eisai, Eli Lilly, Gilead, Jansen, Novartis, Sanofi, Tanabe Mitsubishi, and UCB; and research grants from AbbVie, Mochida, and Takeda. K.F. received grants, consulting fees, speaking fees, and/or honoraria from Chugai Pharmaceutical Co., Ltd., Asahi Kasei Pharma, AstraZeneca, Tsumura & Co., AbbVie, Bristol Myers Squibb, Eisai, Taisho Pharmaceutical, Eli Lilly, Daiichi Sankyo, GlaxoSmithKline, Mitsubishi Tanabe Pharma, Gilead Sciences, Pfizer Inc., Astellas Pharma Inc., Novartis, and Alexion Pharmaceuticals, Inc.

Abbreviations

APC	antigen-presenting cells		
APRIL	A proliferation-inducing ligand		
BAFF/BlyS	B cell activating factor/B lymphocyte stimulator		
BAFF-R	B cell activating factor receptor		
BCMA	B cell maturation antigen		
BDCA2	blood dendritic cell antigen 2		
BEL/PBO	belimumab with placebo		
BEL/RTX	belimumab with rituximab		
BICLA	British Isles Lupus Assessment Group-based Composite Lupus Assessment		
BILAG	British Isles Lupus Assessment Group (index)		
CAR	chimeric antigen receptor		
CD	cluster of differentiation		
CD40L	cluster of differentiation 40 ligand		
CRS	cytokine release syndrome		
CXCL10	cysteine-X-cysteine motif chemokine ligand 10		
DORIS	Definition Of Remission In SLE		
eGFR	estimated glomerular filtration rate		
eQTLs	expression quantitative trait loci		
EULAR	European League Against Rheumatism		
GFR	glomerular filtration rate		
GWAS	genome-wide association studies		
HZ	herpes zoster		
ICANS	immune effector cell-associated neurotoxicity syndrome		
IFN	interferon		
IFNAR	interferon receptor		
IFNAR1	interferon receptor 1		
IGS	interferon gene signature		
Ig	immunoglobulin		
IL	interleukin		
JAK	Janus kinase		
JAK1	Janus kinase 1		
KDIGO	Kidney Disease: Improving Global Outcomes		
LLDAS	Lupus Low Disease Activity State		
LN	lupus nephritis		
pDC	plasmacytoid dendritic cells		
PDGFRA	platelet-derived growth factor receptor alpha		
PGA	Physician's Global Assessment		
RA	rheumatoid arthritis		
SELENA-SLEDAI	Safety of Estrogens in Lupus Erythematosus National Assessment–Systemic Lupus Erythematosus Disease Activity Index		
SLE	systemic lupus erythematosus		

SLEDAI-2K	Systemic Lupus Erythematosus Disease Activity Index 2000	
SRI-4	Systemic Lupus Erythematosus Responder Index	
TACI	transmembrane activator, calcium modulator, and cyclophilin	
IACI	ligand interactor	
TCE	T cell engager	
Treg	regulatory T cell	
TYK2	Tyrosine kinase	

References

- 1. Tsokos, G.C. The immunology of systemic lupus erythematosus. Nat. Immunol. 2024, 25, 1332–1343. [CrossRef] [PubMed]
- Ugarte-Gil, M.F.; Hanly, J.; Urowitz, M.; Gordon, C.; Bae, S.C.; Romero-Diaz, J.; Sanchez-Guerrero, J.; Bernatsky, S.; Clarke, A.E.; Wallace, D.J.; et al. Remission and low disease activity (LDA) prevent damage accrual in patients with systemic lupus erythematosus: Results from the Systemic Lupus International Collaborating Clinics (SLICC) inception cohort. *Ann. Rheum. Dis.* 2022, *81*, 1541–1548. [CrossRef]
- Figueroa-Parra, G.; Cuéllar-Gutiérrez, M.C.; González-Treviño, M.; Sanchez-Rodriguez, A.; Flores-Gouyonnet, J.; Meade-Aguilar, J.A.; Prokop, L.J.; Murad, M.H.; Dall'Era, M.; Rovin, B.H.; et al. Impact of glucocorticoid dose on complete response, serious infections, and mortality during the initial therapy of lupus nephritis: A systematic review and meta-analysis of the control arms of randomized controlled trials. *Arthritis Rheumatol.* 2024, *76*, 1408–1418. [CrossRef] [PubMed]
- 4. Bultink IE, M.; de Vries, F.; van Vollenhoven, R.F.; Lalmohamed, A. Mortality, causes of death and influence of medication use in patients with systemic lupus erythematosus vs matched controls. *Rheumatology* **2021**, *60*, 207–216. [CrossRef] [PubMed]
- Chen, H.-L.; Shen, L.J.; Hsu, P.N.; Shen, C.Y.; Hall, S.A.; Hsiao, F.Y. Cumulative burden of glucocorticoid-related adverse events in patients with systemic lupus erythematosus: Findings from a 12-year longitudinal study. J. Rheumatol. 2018, 45, 83–89. [CrossRef]
- Usiskin, I.M.; Kyttaris, V.C. Estimating glucocorticoid-related morbidity in lupus nephritis using the glucocorticoid toxicity index. Lupus 2023, 32, 565–570. [CrossRef]
- Fanouriakis, A.; Kostopoulou, M.; Andersen, J.; Aringer, M.; Arnaud, L.; Bae, S.C.; Boletis, J.; Bruce, I.N.; Cervera, R.; Doria, A.; et al. EULAR recommendations for the management of systemic lupus erythematosus: 2023 update. *Ann. Rheum. Dis.* 2024, *83*, 15–29. [CrossRef]
- 8. Kidney Disease: Improving Global Outcomes (KDIGO) Lupus Nephritis Work Group. KDIGO 2024 clinical practice guideline for the management of LUPUS NEPHRITIS. *Kidney Int.* 2024, 105, S1–S69. [CrossRef]
- Navarra, S.V.; Guzmán, R.M.; Gallacher, A.E.; Hall, S.; Levy, R.A.; Jimenez, R.E.; Li, E.K.; Thomas, M.; Kim, H.Y.; Leónm, M.G.; et al. Efficacy and safety of belimumab in patients with active systemic lupus erythematosus: A randomised, placebo-controlled, phase 3 trial. *Lancet* 2011, 377, 721–731. [CrossRef]
- Furie, R.; Petri, M.; Zamani, O.; Cervera, R.; Wallace, D.J.; Tegzová, D.; Sanchez-Guerrero, J.; Schwarting, A.; Merrill, J.T.; Chatham, W.W.; et al. A phase III, randomized, placebo-controlled study of belimumab, a monoclonal antibody that inhibits B lymphocyte stimulator, in patients with systemic lupus erythematosus. *Arthritis Rheum.* 2011, *63*, 3918–3930. [CrossRef]
- Furie, R.A.; Petri, M.A.; Wallace, D.J.; Ginzler, E.M.; Merrill, J.T.; Stohl, W.; Chatham, W.W.; Strand, V.; Weinstein, A.; Chevrier, M.R.; et al. Novel evidence-based systemic lupus erythematosus responder index. *Arthritis Rheum.* 2009, 61, 1143–1151. [CrossRef] [PubMed]
- van Vollenhoven, R.F.; Petri, M.A.; Cervera, R.; Roth, D.A.; Ji, B.N.; Kleoudis, C.S.; Zhong, Z.J.; Freimuth, W. Belimumab in the treatment of systemic lupus erythematosus: High disease activity predictors of response. *Ann. Rheum. Dis.* 2012, *71*, 1343–1349.
 [CrossRef] [PubMed]
- 13. Furie, R.; Rovin, B.H.; Houssiau, F.; Malvar, A.; Teng, Y.K.O.; Contreras, G.; Amoura, Z.; Yu, X.; Mok, C.C.; Santiago, M.B.; et al. Two-year, randomized, controlled trial of belimumab in lupus nephritis. *N. Engl. J. Med.* **2020**, *383*, 1117–1128. [CrossRef]
- 14. Rovin, B.H.; Furie, R.; Teng, Y.K.O.; Contreras, G.; Malvar, A.; Yu, X.; Ji, B.; Green, Y.; Gonzalez-Rivera, T.; Bass, D.; et al. A secondary analysis of the Belimumab International Study in Lupus Nephritis trial examined effects of belimumab on kidney outcomes and preservation of kidney function in patients with lupus nephritis. *Kidney Int.* **2022**, *101*, 403–413. [CrossRef]
- Hunnicutt, J.N.; Fairburn-Beech, J.; Georgiou, M.E.; Richards, A.; Gregan, Y.I.; Quasny, H.; Chauhan, D. Evaluating disease control following belimumab treatment in patients with SLE enrolled in the US OBSErve study. *Lupus Sci. Med.* 2022, 9, e000710. [CrossRef]
- 16. Singh, J.A.; Shah, N.P.; Mudano, A.S. Belimumab for systemic lupus erythematosus. *Cochrane Database Syst. Rev.* 2021, 2, CD010668. [CrossRef]
- 17. Gatto, M.; Saccon, F.; Zen, M.; Regola, F.; Fredi, M.; Andreoli, L.; Tincani, A.; Urban, M.L.; Emmi, G.; Ceccarelli, F.; et al. Early disease and low baseline damage as predictors of response to belimumab in patients with systemic lupus erythematosus in a real-life setting. *Arthritis Rheumatol.* **2020**, *72*, 1314–1324. [CrossRef]

- 18. Tanaka, Y.; Kusuda, M.; Yamaguchi, Y. Interferons and systemic lupus erythematosus: Pathogenesis, clinical features, and treatments in interferon-driven disease. *Mod. Rheumatol.* **2023**, *33*, 857–867. [CrossRef]
- 19. Tanaka, Y.; Tummala, R. Anifrolumab, a monoclonal antibody to the type I interferon receptor subunit 1, for the treatment of systemic lupus erythematosus: An overview from clinical trials. *Mod. Rheumatol.* **2021**, *31*, 1–12. [CrossRef]
- 20. Furie, R.A.; Morand, E.F.; Bruce, I.N.; Manzi, S.; Kalunian, K.C.; Vital, E.M.; Lawrence Ford, T.; Gupta, R.; Hiepe, F.; Santiago, M.; et al. Type I interferon inhibitor anifrolumab in active systemic lupus erythematosus (TULIP-1): A randomised, controlled, phase 3 trial. *Lancet Rheumatol.* **2019**, *1*, e208–e219. [CrossRef]
- 21. Morand, E.F.; Furie, R.; Tanaka, Y.; Bruce, I.N.; Askanase, A.D.; Richez, C.; Bae, S.C.; Brohawn, P.Z.; Pineda, L.; Berglind, A.; et al. Trial of anifrolumab in active systemic lupus erythematosus. *N. Engl. J. Med.* **2020**, *382*, 211–221. [CrossRef] [PubMed]
- 22. Murphy, C.-L.; Yee, C.-S.; Gordon, C.; Isenberg, D. From BILAG to BILAG-based combined lupus assessment-30 years on. *Rheumatology* **2016**, *55*, 1357–1363. [CrossRef] [PubMed]
- 23. Morand, E.F.; Abreu, G.; Furie, R.A.; Golder, V.; Tummala, R. Lupus low disease activity state attainment in the phase 3 TULIP trials of anifrolumab in active systemic lupus erythematosus. *Ann. Rheum. Dis.* **2023**, *82*, 639–645. [CrossRef] [PubMed]
- 24. Merrill, J.T.; Furie, R.; Werth, V.P.; Khamashta, M.; Drappa, J.; Wang, L.; Illei, G.; Tummala, R. Anifrolumab effects on rash and arthritis: Impact of the type I interferon gene signature in the phase IIb MUSE study in patients with systemic lupus erythematosus. *Lupus Sci. Med.* **2018**, *5*, e000284. [CrossRef]
- Chatham, W.W.; Furie, R.; Saxena, A.; Brohawn, P.; Schwetje, E.; Abreu, G.; Tummala, R. Long-term safety and efficacy of anifrolumab in adults with systemic lupus erythematosus: Results of a phase II open-label extension study. *Arthritis Rheumatol.* 2021, 73, 816–825. [CrossRef]
- Kalunian, K.C.; Furie, R.; Morand, E.F.; Bruce, I.N.; Manzi, S.; Tanaka, Y.; Winthrop, K.; Hupka, I.; Zhang, L.J.; Werther, S.; et al. A randomized, placebo-controlled phase III extension trial of the long-term safety and tolerability of anifrolumab in active systemic lupus erythematosus. *Arthritis Rheumatol.* 2023, 75, 253–265. [CrossRef]
- Jayne, D.; Rovin, B.; Mysler, E.F.; Furie, R.A.; Houssiau, F.A.; Trasieva, T.; Knagenhjelm, J.; Schwetje, E.; Chia, Y.L.; Tummala, R.; et al. Phase II randomised trial of type I interferon inhibitor anifrolumab in patients with active lupus nephritis. *Ann. Rheum. Dis.* 2022, *81*, 496–506. [CrossRef]
- Merrill, J.T.; Neuwelt, C.M.; Wallace, D.J.; Shanahan, J.C.; Latinis, K.M.; Oates, J.C.; Utset, T.O.; Gordon, C.; Isenberg, D.A.; Hsieh, H.J.; et al. Efficacy and safety of rituximab in moderately-to-severely active systemic lupus erythematosus: The randomized, double-blind, phase II/III systemic lupus erythematosus evaluation of rituximab trial. *Arthritis Rheum.* 2010, 62, 222–233. [CrossRef]
- 29. Rovin, B.H.; Furie, R.; Latinis, K.; Looney, R.J.; Fervenza, F.C.; Sanchez-Guerrero, J.; Maciuca, R.; Zhang, D.; Garg, J.P.; Brunetta, P.; et al. Efficacy and safety of rituximab in patients with active proliferative lupus nephritis: The Lupus Nephritis Assessment with Rituximab study. *Arthritis Rheum.* **2012**, *64*, 1215–1226. [CrossRef]
- McCarthy, E.M.; Sutton, E.; Nesbit, S.; White, J.; Parker, B.; Jayne, D.; Griffiths, B.; Isenberg, D.A.; Rahman, A.; Gordon, C.; et al. Short-term efficacy and safety of rituximab therapy in refractory systemic lupus erythematosus: Results from the British Isles Lupus Assessment Group Biologics Register. *Rheumatology* 2018, *57*, 470–479. [CrossRef]
- 31. Terrier, B.; Amoura, Z.; Ravaud, P.; Hachulla, E.; Jouenne, R.; Combe, B.; Bonnet, C.; Cacoub, P.; Cantagrel, A.; de Bandt, M.; et al. Safety and efficacy of rituximab in systemic lupus erythematosus: Results from 136 patients from the French AutoImmunity and Rituximab registry. *Arthritis Rheum.* **2010**, *62*, 2458–2466. [CrossRef]
- 32. Serris, A.; Amoura, Z.; Canouï-Poitrine, F.; Terrier, B.; Hachulla, E.; Costedoat-Chalumeau, N.; Papo, T.; Lambotte, O.; Saadoun, D.; Hié, M.; et al. Efficacy and safety of rituximab for systemic lupus erythematosus-associated immune cytopenias: A multicenter retrospective cohort study of 71 adults. *Am. J. Hematol.* 2018, 93, 424–429. [CrossRef] [PubMed]
- 33. Atisha-Fregoso, Y.; Malkiel, S.; Harris, K.M.; Byron, M.; Ding, L.; Kanaparthi, S.; Barry, W.T.; Gao, W.; Ryker, K.; Tosta, P.; et al. Phase II randomized trial of rituximab plus cyclophosphamide followed by belimumab for the treatment of lupus nephritis. *Arthritis Rheumatol.* 2021, 73, 121–131. [CrossRef] [PubMed]
- 34. Shipa, M.; Santos, L.R.; Nguyen, D.X.; Embleton-Thirsk, A.; Parvaz, M.; Heptinstall, L.L.; Pepper, R.J.; Isenberg, D.A.; Gordon, C.; Ehrenstein, M.R. Identification of biomarkers to stratify response to B-cell-targeted therapies in systemic lupus erythematosus: An exploratory analysis of a randomised controlled trial. *Lancet Rheumatol.* **2023**, *5*, e24–e35. [CrossRef] [PubMed]
- 35. Aranow, C.; Allaart, C.F.; Amoura, Z.; Bruce, I.N.; Cagnoli, P.C.; Chatham, W.W.; Clark, K.L.; Furie, R.; Groark, J.; Urowitz, M.B.; et al. Efficacy and safety of sequential therapy with subcutaneous belimumab and one cycle of rituximab in patients with systemic lupus erythematosus: The phase 3, randomised, placebo-controlled BLISS-BELIEVE study. *Ann. Rheum. Dis.* 2024, *83*, 1502–1512. [CrossRef]
- 36. Rovin, B.H.; Teng, Y.K.O.; Ginzler, E.M.; Arriens, C.; Caster, D.J.; Romero-Diaz, J.; Gibson, K.; Kaplan, J.; Lisk, L.; Navarra, S.; et al. Efficacy and safety of voclosporin versus placebo for lupus nephritis (AURORA 1): A double-blind, randomised, multicentre, placebo-controlled, phase 3 trial. *Lancet* 2021, 397, 2070–2080. [CrossRef]

- 37. Saxena, A.; Ginzler, E.M.; Gibson, K.; Satirapoj, B.; Santillán, A.E.Z.; Levchenko, O.; Navarra, S.; Atsumi, T.; Yasuda, S.; Chavez-Perez, N.N.; et al. Safety and efficacy of long-term voclosporin treatment for lupus nephritis in the phase 3 AURORA 2 clinical trial. *Arthritis Rheumatol.* **2024**, *76*, 59–67. [CrossRef]
- Reddy, V.; Klein, C.; Isenberg, D.A.; Glennie, M.J.; Cambridge, G.; Cragg, M.S.; Leandro, M.J. Obinutuzumab induces superior B-cell cytotoxicity to rituximab in rheumatoid arthritis and systemic lupus erythematosus patient samples. *Rheumatology* 2017, 56, 1227–1237. [CrossRef]
- Furie, R.A.; Aroca, G.; Cascino, M.D.; Garg, J.P.; Rovin, B.H.; Alvarez, A.; Fragoso-Loyo, H.; Zuta-Santillan, E.; Schindler, T.; Brunetta, P.; et al. B-cell depletion with obinutuzumab for the treatment of proliferative lupus nephritis: A randomised, double-blind, placebo-controlled trial. *Ann. Rheum. Dis.* 2022, *81*, 100–107. [CrossRef]
- Arnold, J.; Dass, S.; Twigg, S.; Jones, C.H.; Rhodes, B.; Hewins, P.; Chakravorty, M.; Courtney, P.; Ehrenstein, M.; Md Yusof, M.Y.; et al. Efficacy and safety of obinutuzumab in systemic lupus erythematosus patients with secondary non-response to rituximab. *Rheumatology* 2022, *61*, 4905–4909. [CrossRef]
- 41. Boumpas, D.T.; Furie, R.; Manzi, S.; Illei, G.G.; Wallace, D.J.; Balow, J.E.; Vaishnaw, A. A short course of BG9588 (anti–CD40 ligand antibody) improves serologic activity and decreases hematuria in patients with proliferative lupus glomerulonephritis. *Arthritis Rheum.* **2003**, *48*, 719–727. [CrossRef] [PubMed]
- 42. Furie, R.A.; Bruce, I.N.; Dörner, T.; Leon, M.G.; Leszczyński, P.; Urowitz, M.; Haier, B.; Jimenez, T.; Brittain, C.; Liu, J.; et al. Phase 2, randomized, placebo-controlled trial of dapirolizumab pegol in patients with moderate-to-severe active systemic lupus erythematosus. *Rheumatology* **2021**, *60*, 5397–5407. [CrossRef] [PubMed]
- 43. Megan Clowse, David Isenberg, Joan Merrill, Thomas Dörner, Michelle Petri, Edward Vital, Eric Morand, Teri Jimenez, Stephen Brookes, Janine Gaiha-Rohrbach, Christophe Martin, Annette Nelde and Christian Stach. Dapirolizumab Pegol Demonstrated Significant Improvement in Systemic Lupus Erythematosus Disease Activity: Efficacy and Safety Results of a Phase 3 Trial. ACR Convergence 2024. 2024. Available online: https://acrabstracts.org/abstract/dapirolizumab-pegol-demonstrated-significantimprovement-in-systemic-lupus-erythematosus-disease-activity-efficacy-and-safety-results-of-a-phase-3-trial/ (accessed on 21 December 2024).
- Wu, D.; Li, J.; Xu, D.; Merrill, J.T.; van Vollenhoven, R.F.; Liu, Y.; Hu, J.; Li, Y.; Li, F.; Huang, C.; et al. Telitacicept in patients with active systemic lupus erythematosus: Results of a phase 2b, randomised, double-blind, placebo-controlled trial. *Ann. Rheum. Dis.* 2024, *83*, 475–487.
- Efficacy and Immunological Evaluation of Telitacicept and Low Dose IL2 in the Treatment of Systemic Lupus Erythematosus: A Randomise Prospective Study. *Clinicaltrials.gov*. Available online: https://clinicaltrials.gov/study/NCT05339217 (accessed on 16 January 2025).
- Furie, R.A.; van Vollenhoven, R.F.; Kalunian, K.; Navarra, S.; Romero-Diaz, J.; Werth, V.P.; Huang, X.; Clark, G.; Carroll, H.; Meyers, A.; et al. Trial of anti-BDCA2 antibody litifilimab for systemic lupus erythematosus. *N. Engl. J. Med.* 2022, 387, 894–904. [CrossRef] [PubMed]
- 47. Merrill, J.T.; Tanaka, Y.; D'Cruz, D.; Vila-Rivera, K.; Siri, D.; Zeng, X.; Saxena, A.; Aringer, M.; D'Silva, K.M.; Cheng, L.; et al. Efficacy and safety of upadacitinib or elsubrutinib alone or in combination for patients with systemic lupus erythematosus: A phase 2 randomized controlled trial. *Arthritis Rheumatol.* **2024**, *76*, 1518–1529.
- 48. Morand, E.; Pike, M.; Merrill, J.T.; van Vollenhoven, R.; Werth, V.P.; Hobar, C.; Delev, N.; Shah, V.; Sharkey, B.; Wegman, T.; et al. Deucravacitinib, a tyrosine kinase 2 inhibitor, in systemic lupus erythematosus: A phase II, randomized, double-blind, placebo-controlled trial. *Arthritis Rheumatol.* 2023, 75, 242–252. [CrossRef]
- 49. Arriens, C.; Askanase, A.; Furie, R.; Morand, E.F.; Van Vollenhoven, R.; Connors, K.; Davey, M.; Delev, N.; Shah, V.; Stevens, A.; et al. Ab0530 design of 2 phase 3, double-blind, placebo-controlled, global trials of deucravacitinib, an oral, selective, allosteric tyrosine kinase 2 (tyk2) inhibitor, in patients with active systemic lupus erythematosus. In *Scientific Abstracts*; BMJ Publishing Group Ltd.: London, UK; European League Against Rheumatism: Zürich, Switzerland, 2023; Volume 82.
- He, J.; Zhang, R.; Shao, M.; Zhao, X.; Miao, M.; Chen, J.; Liu, J.; Zhang, X.; Zhang, X.; Jin, Y.; et al. Efficacy and safety of low-dose IL-2 in the treatment of systemic lupus erythematosus: A randomised, double-blind, placebo-controlled trial. *Ann. Rheum. Dis.* 2020, *79*, 141–149.
- 51. Rosenzwajg, M.; Lorenzon, R.; Cacoub, P.; Pham, H.P.; Pitoiset, F.; El Soufi, K.; RIbet, C.; Bernard, C.; Aractingi, S.; Banneville, B.; et al. Immunological and clinical effects of low-dose interleukin-2 across 11 autoimmune diseases in a single, open clinical trial. *Ann. Rheum. Dis.* 2019, *78*, 209–217. [CrossRef] [PubMed]
- 52. A Prospective, Single-Center Study of Comparing the Efficacy and Safety of Human Umbilical Cord Mesenchymal Stem Cells and Low-dose IL-2 in the Treatment of Lupus Nephritis. *Clinicaltrials.gov*. Available online: https://clinicaltrials.gov/study/ NCT05631717 (accessed on 16 January 2025).
- 53. van Vollenhoven, R.F.; Hahn, B.H.; Tsokos, G.C.; Wagner, C.L.; Lipsky, P.; Touma, Z.; Werth, V.P.; Gordon, R.M.; Zhou, B.; Hsu, B.; et al. Efficacy and safety of ustekinumab, an IL-12 and IL-23 inhibitor, in patients with active systemic lupus erythematosus: Results of a multicentre, double-blind, phase 2, randomised, controlled study. *Lancet* **2018**, *392*, 1330–1339. [CrossRef]

- 54. van Vollenhoven, R.F.; Kalunian, K.C.; Dörner, T.; Hahn, B.H.; Tanaka, Y.; Gordon, R.M.; Shu, C.; Fei, K.; Gao, S.; Seridi, L.; et al. Phase 3, multicentre, randomised, placebo-controlled study evaluating the efficacy and safety of ustekinumab in patients with systemic lupus erythematosus. *Ann. Rheum. Dis.* **2022**, *81*, 1556–1563.
- 55. Costa, R.; Antunes, P.; Salvador, P.; Oliveira, P.; Marinho, A. Secukinumab on refractory lupus nephritis. Cureus 2021, 13, e17198.
- 56. Sato, K.; Aizaki, Y.; Yoshida, Y.; Mimura, T. Treatment of psoriatic arthritis complicated by systemic lupus erythematosus with the IL-17 blocker secukinumab and an analysis of the serum cytokine profile. *Mod. Rheumatol. Case Rep.* 2020, *4*, 181–185. [CrossRef] [PubMed]
- 57. A Pilot Study to Assess the Safety and Efficacy of Secukinumab in Alleviating Symptoms of Discoid Lupus Erythematosus. Clinicaltrials.gov. Available online: https://clinicaltrials.gov/study/NCT03866317 (accessed on 16 January 2025).
- 58. A Two-year, Phase III Randomized, Double-Blind, Parallel-Group, Placebo-Controlled Trial to Evaluate the Safety, Efficacy, and Tolerability of 300 mg s.c. Secukinumab Versus Placebo, in Combination With SoC Therapy, in Patients With Active Lupus Nephritis. Clinicaltrials.gov. Available online: https://clinicaltrials.gov/study/NCT04181762?cond=Systemic%20Lupus%20 Erythematosus&intr=Secukinumab&rank=3 (accessed on 16 January 2025).
- A Three-year, Open-Label Extension Study of Subcutaneous Secukinumab to Evaluate the Long-term Efficacy, Safety and Tolerability in Patients with Active Lupus Nephritis. Clinicaltrials.gov. Available online: https://clinicaltrials.gov/study/NCT05232864 (accessed on 16 January 2025).
- 60. Koller-Smith, L.; Oakley, S. Secukinumab-induced systemic lupus erythematosus occurring in a patient with ankylosing spondylitis. *Rheumatology* **2022**, *61*, e146–e147. [CrossRef] [PubMed]
- 61. Kaler, J.; Kaeley, G.S. Secukinumab-induced lupus erythematosus: A case report and literature review. *J. Clin. Rheumatol.* 2021, 27, S753–S754. [CrossRef] [PubMed]
- 62. Schett, G.; Mackensen, A.; Mougiakakos, D. CAR T-cell therapy in autoimmune diseases. Lancet 2023, 402, 2034–2044. [CrossRef]
- 63. Schett, G.; Müller, F.; Taubmann, J.; Mackensen, A.; Wang, W.; Furie, R.A.; Gold, R.; Haghikia, A.; Merkel, P.A.; Caricchio, R.; et al. Advancements and challenges in CAR T cell therapy in autoimmune diseases. *Nat. Rev. Rheumatol.* **2024**, *20*, 531–544. [CrossRef]
- 64. Mougiakakos, D.; Krönke, G.; Völkl, S.; Kretschmann, S.; Aigner, M.; Kharboutli, S.; Böltz, S.; Manger, B.; Mackensen, A.; Schett, G. CD19-targeted CAR T cells in refractory systemic lupus erythematosus. *N. Engl. J. Med.* **2021**, *385*, 567–569. [CrossRef]
- 65. Mackensen, A.; Müller, F.; Mougiakakos, D.; Böltz, S.; Wilhelm, A.; Aigner, M.; Völkl, S.; Simon, D.; Kleyer, A.; Munoz, L.; et al. Anti-CD19 CAR T cell therapy for refractory systemic lupus erythematosus. *Nat. Med.* **2022**, *28*, 2124–2132. [CrossRef]
- 66. Müller, F.; Taubmann, J.; Bucci, L.; Wilhelm, A.; Bergmann, C.; Völkl, S.; Aigner, M.; Rothe, T.; Minopoulou, I.; Tur, C.; et al. CD19 CAR T-cell therapy in autoimmune disease—A case series with follow-up. *N. Engl. J. Med.* **2024**, 390, 687–700. [CrossRef]
- 67. Wang, W.; He, S.; Zhang, W.; Zhang, H.; DeStefano, V.M.; Wada, M.; Pinz, K.; Deener, G.; Shah, D.; Hagag, N.; et al. BCMA-CD19 compound CAR T cells for systemic lupus erythematosus: A phase 1 open-label clinical trial. *Ann. Rheum. Dis.* **2024**, *83*, 1304–1314. [CrossRef]
- 68. Hagen, M.; Bucci, L.; Böltz, S.; Nöthling, D.M.; Rothe, T.; Anoshkin, K.; Raimondo, M.G.; Tur, C.; Wirsching, A.; Wacker, J.; et al. BCMA-targeted T-cell-engager therapy for autoimmune disease. *N. Engl. J. Med.* **2024**, *391*, 867–869. [CrossRef]
- 69. Alexander, T.; Krönke, J.; Cheng, Q.; Keller, U.; Krönke, G. Teclistamab-induced remission in refractory systemic lupus erythematosus. N. Engl. J. Med. 2024, 391, 864–866. [CrossRef] [PubMed]
- 70. Moreau, P.; Garfall, A.L.; van de Donk, N.W.C.J.; Nahi, H.; San-Miguel, J.F.; Oriol, A.; Nooka, A.K.; Martin, T.; Rosinol, L.; Chari, A.; et al. Teclistamab in relapsed or refractory multiple myeloma. *N. Engl. J. Med.* **2022**, *387*, 495–505. [CrossRef] [PubMed]
- 71. Moore, D.C.; Granger, K.; Hill, H.; Karabinos, A.; Davis, J.A. Elranatamab vs. teclistamab: Battle of the BCMA bispecifics in relapsed/refractory multiple myeloma. *Expert Rev. Hematol.* **2024**, *17*, 197–200. [CrossRef] [PubMed]
- 72. Bucci, L.; Hagen, M.; Rothe, T.; Raimondo, M.G.; Fagni, F.; Tur, C.; Wirsching, A.; Wacker, J.; Wilhelm, A.; Auger, J.P.; et al. Bispecific T cell engager therapy for refractory rheumatoid arthritis. *Nat. Med.* **2024**, *30*, 1593–1601. [CrossRef]
- 73. A Study of CLN-978, a Subcutaneously Administered CD19-Directed T Cell Engager, in Subjects With Systemic Lupus Erythematosus. Clinicaltrials.gov. Available online: https://clinicaltrials.gov/study/NCT06613360 (accessed on 16 December 2024).
- 74. Meetze, K.; Mehta, N.K.; Li, B.; Michaelson, J.S.; Baeuerle, P.A. CLN-978, a novel half-life extended CD19/CD3/HSA-specific T cell-engaging antibody construct with potent activity against B-cell malignancies with low CD19 expression. *J. Immunother. Cancer* 2023, *11*, e007398. [CrossRef]
- 75. Fasano, S.; Milone, A.; Nicoletti, G.F.; Isenberg, D.A.; Ciccia, F. Precision medicine in systemic lupus erythematosus. *Nat. Rev. Rheumatol.* **2023**, *19*, 331–342. [CrossRef]
- 76. Hom, G.; Graham, R.R.; Modrek, B.; Taylor, K.E.; Ortmann, W.; Garnier, S.; Lee, A.T.; Chung, S.A.; Ferreira, R.C.; Pant, P.V.; et al. Association of systemic lupus erythematosus with C8orf13-BLK and ITGAM-ITGAX. N. Engl. J. Med. 2008, 358, 900–909. [CrossRef]
- Kozyrev, S.V.; Abelson, A.K.; Wojcik, J.; Zaghlool, A.; Linga Reddy, M.V.; Sanchez, E.; Gunnarsson, I.; Svenungsson, E.; Sturfelt, G.; Jönsen, A.; et al. Functional variants in the B-cell gene BANK1 are associated with systemic lupus erythematosus. *Nat. Genet.* 2008, 40, 211–216. [CrossRef]

- 78. Yasutomo, K.; Horiuchi, T.; Kagami, S.; Tsukamoto, H.; Hashimura, C.; Urushihara, M.; Kuroda, Y. Mutation of DNASE1 in people with systemic lupus erythematosus. *Nat. Genet.* **2001**, *28*, 313–314. [CrossRef]
- Al-Mayouf, S.M.; Sunker, A.; Abdwani, R.; Abrawi, S.A.; Almurshedi, F.; Alhashmi, N.; Al Sonbul, A.; Sewairi, W.; Qari, A.; Abdallah, E.; et al. Loss-of-function variant in DNASE1L3 causes a familial form of systemic lupus erythematosus. *Nat. Genet.* 2011, 43, 1186–1188. [CrossRef]
- Chung, S.A.; Brown, E.E.; Williams, A.H.; Ramos, P.S.; Berthier, C.C.; Bhangale, T.; Alarcon-Riquelme, M.E.; Behrens, T.W.; Criswell, L.A.; Graham, D.C.; et al. Lupus nephritis susceptibility loci in women with systemic lupus erythematosus. *J. Am. Soc. Nephrol.* 2014, 25, 2859–2870. [CrossRef] [PubMed]
- Feusi, E.; Sun, L.; Sibalic, A.; Beck-Schimmer, B.; Oertli, B.; Wüthrich, R.P. Enhanced Hyaluronan Synthesis in the MRL-Fas^{lpr} Kidney: Role of Cytokines. *Nephron* 1999, 83, 66–73. [CrossRef] [PubMed]
- Järvinen, T.M.; Hellquist, A.; Koskenmies, S.; Einarsdottir, E.; Panelius, J.; Hasan, T.; Julkunen, H.; Padyukov, L.; Kvarnström, M.; Wahren-Herlenius, M.; et al. Polymorphisms of the ITGAM gene confer higher risk of discoid cutaneous than of systemic lupus erythematosus. *PLoS ONE* 2010, 5, e14212. [CrossRef] [PubMed]
- Taylor, K.E.; Chung, S.A.; Graham, R.R.; Ortmann, W.A.; Lee, A.T.; Langefeld, C.D.; Jacob, C.O.; Kamboh, M.I.; Alarcón-Riquelme, M.E.; Tsao, B.P.; et al. Risk alleles for systemic lupus erythematosus in a large case-control collection and associations with clinical subphenotypes. *PLoS Genet.* 2011, 7, e1001311. [CrossRef]
- 84. Ho, R.C.; Ong, H.; Thiaghu, C.; Lu, Y.; Ho, C.S.; Zhang, M.W. Genetic variants that are associated with neuropsychiatric systemic lupus erythematosus. *J. Rheumatol.* **2016**, *43*, 541–551. [CrossRef]
- Perez, R.K.; Gordon, M.G.; Subramaniam, M.; Kim, M.C.; Hartoularos, G.C.; Targ, S.; Sun, Y.; Ogorodnikov, A.; Bueno, R.; Lu, A.; et al. Single-cell RNA-seq reveals cell type-specific molecular and genetic associations to lupus. *Science* 2022, 376, eabf1970.
 [CrossRef]
- 86. Adams, D.E.; Shao, W.H. Epigenetic alterations in immune cells of systemic lupus erythematosus and therapeutic implications. *Cells* **2022**, *11*, 506. [CrossRef]
- 87. Perez-Hernandez, J.; Forner, M.J.; Pinto, C.; Chaves, F.J.; Cortes, R.; Redon, J. Increased urinary exosomal MicroRNAs in patients with systemic lupus erythematosus. *PLoS ONE* **2015**, *10*, e0138618. [CrossRef]
- 88. Zhang, R.; Li, Y.; Pan, B.; Li, Y.; Liu, A.; Li, X. Increased expression of hub gene CXCL10 in peripheral blood mononuclear cells of patients with systemic lupus erythematosus. *Exp. Ther. Med.* **2019**, *18*, 4067–4075. [CrossRef]
- 89. Narumi, S.; Takeuchi, T.; Kobayashi, Y.; Konishi, K. Serum levels of ifn-inducible PROTEIN-10 relating to the activity of systemic lupus erythematosus. *Cytokine* **2000**, *12*, 1561–1565. [CrossRef]
- 90. Capecchi, R.; Puxeddu, I.; Pratesi, F.; Migliorini, P. New biomarkers in SLE: From bench to bedside. *Rheumatology* **2020**, *59*, v12–v18. [CrossRef] [PubMed]
- Fragoso-Loyo, H.; Richaud-Patin, Y.; Orozco-Narváez, A.; Dávila-Maldonado, L.; Atisha-Fregoso, Y.; Llorente, L.; Sánchez-Guerrero, J. Interleukin-6 and chemokines in the neuropsychiatric manifestations of systemic lupus erythematosus. *Arthritis Rheum.* 2007, 56, 1242–1250. [CrossRef] [PubMed]
- 92. van den Hoogen, L.L.; van Roon, J.A.G.; Mertens, J.S.; Wienke, J.; Lopes, A.P.; de Jager, W.; Rossato, M.; Pandit, A.; Wichers, C.G.K.; van Wijk, F.; et al. Galectin-9 is an easy to measure biomarker for the interferon signature in systemic lupus erythematosus and antiphospholipid syndrome. *Ann. Rheum. Dis.* 2018, 77, 1810–1814. [CrossRef] [PubMed]
- 93. Oliveira, J.J.; Karrar, S.; Rainbow, D.B.; Pinder, C.L.; Clarke, P.; Rubio García, A.; Al-Assar, O.; Burling, K.; Morris, S.; Stratton, R.; et al. The plasma biomarker soluble SIGLEC-1 is associated with the type I interferon transcriptional signature, ethnic background and renal disease in systemic lupus erythematosus. *Arthritis Res. Ther.* **2018**, *20*, 152. [CrossRef]
- Zollars, E.; Bienkowska, J.; Czerkowicz, J.; Allaire, N.; Ranger, A.M.; Magder, L.; Petri, M. BAFF (B cell activating factor) transcript level in peripheral blood of patients with SLE is associated with same-day disease activity as well as global activity over the next year. *Lupus Sci. Med.* 2015, 2, e000063. [CrossRef]
- 95. Lee, Y.H.; Song, G.G. Urinary MCP-1 as a biomarker for lupus nephritis: A meta-analysis. Z. Rheumatol. 2017, 76, 357–363. [CrossRef]
- 96. Xuejing, Z.; Jiazhen, T.; Jun, L.; Xiangqing, X.; Shuguang, Y.; Fuyou, L. Urinary TWEAK level as a marker of lupus nephritis activity in 46 cases. *J. Biomed. Biotechnol.* **2012**, 2012, 359647. [CrossRef]
- 97. Rubinstein, T.; Pitashny, M.; Levine, B.; Schwartz, N.; Schwartzman, J.; Weinstein, E.; Pego-Reigosa, J.M.; Lu, T.Y.; Isenberg, D.; Rahman, A.; et al. Urinary neutrophil gelatinase-associated lipocalin as a novel biomarker for disease activity in lupus nephritis. *Rheumatology* **2010**, *49*, 960–971. [CrossRef]
- Sawada, T.; Kurano, M.; Shirai, H.; Iwasaki, Y.; Tahara, K.; Hayashi, H.; Igarashi, K.; Fujio, K.; Aoki, J.; Yatomi, Y. Serum phosphatidylserine-specific phospholipase A1 as a novel biomarker for monitoring systemic lupus erythematosus disease activity. *Int. J. Rheum. Dis.* 2019, 22, 2059–2066. [CrossRef]
- 99. Nakano, M.; Ota, M.; Takeshima, Y.; Iwasaki, Y.; Hatano, H.; Nagafuchi, Y.; Itamiya, T.; Maeda, J.; Yoshida, R.; Yamada, S.; et al. Distinct transcriptome architectures underlying lupus establishment and exacerbation. *Cell* **2022**, *185*, 3375–3389.e21. [CrossRef]

- 100. Robinson, G.A.; Peng, J.; Dönnes, P.; Coelewij, L.; Naja, M.; Radziszewska, A.; Wincup, C.; Peckham, H.; Isenberg, D.A.; Ioannou, Y.; et al. Disease-associated and patient-specific immune cell signatures in juvenile-onset systemic lupus erythematosus: Patient stratification using a machine-learning approach. *Lancet Rheumatol.* 2020, *2*, e485–e496. [CrossRef] [PubMed]
- 101. Ota, M.; Nagafuchi, Y.; Hatano, H.; Ishigaki, K.; Terao, C.; Takeshima, Y.; Yanaoka, H.; Kobayashi, S.; Okubo, M.; Shirai, H.; et al. Dynamic landscape of immune cell-specific gene regulation in immune-mediated diseases. *Cell* 2021, *184*, 3006–3021.e17. [CrossRef] [PubMed]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.